Author's response to reviews

Title: Threonyl-tRNA Synthetase Overexpression Correlates with Angiogenic Markers and Progression of Human Ovarian Cancer

Authors:

Theresa L Wellman (theresa.wellman@uvm.edu)
Midori Eckenstein (midori.eckenstein@gmail.com)
Cheung Wong (cheung.wong@vtmednet.org)
Mercedes Rincon (mercedes.rincon@uvm.edu)
Takamura Ashikaga (takamura.ashikaga@uvm.edu)
Sharon L Mount (sharon.mount@vtmednet.org)
Christopher S Francklyn (christopher.francklyn@uvm.edu)
Karen M Lounsbury (karen.lounsbury@uvm.edu)

Version: 4 Date: 5 August 2014

Author's response to reviews: see over
August 5, 2014

MS: 1924267027127993
Threonyl-tRNA Synthetase Overexpression Correlates with Angiogenesis and Progression of Human Ovarian Cancer Theresa L Wellman, Midori Eckenstein, Cheung Wong, Mercedes Rincon, Takamura Ashikaga, Sharon L Mount, Christopher S Francklyn and Karen M Lounsbury

Regarding the publication of the manuscript we have addressed the following concern.

**Requesting consent statement:**

Please state in the Methods section whether written informed consent for participation in the study was obtained from participants or, where participants are children, a parent or guardian.

*We have added an additional sentence in the methods indicating that all patients signed written consent forms.*

We have checked for journal format, but please let us know if there are any additional edits needed for publication.

Sincerely,

Karen M. Lounsbury, Ph.D.
Corresponding Author
University of Vermont, College of Medicine
E-mail: Karen.Lounsbury@uvm.edu
Tel: (802)656-1319